Author's response to reviews

Title: Angiographically borderline left main coronary artery lesions: correlation of transthoracic Doppler echocardiography and intravascular ultrasound: a pilot study

Authors:

Zoltán Ruzsa (zruzsa25@gmail.com)
Attila Pálinkás (palinkasa@hotmail.com)
Tamás Forster (forster@in2nd.szote.u-szeged.hu)
Imre Ungi (zruzsa25@gmail.com)
Albert Varga (vargaa@in2nd.szote.u-szeged.hu)

Version: 9 Date: 7 June 2011

Author's response to reviews:

Dear Editor,

Please find enclosed our revised manuscript entitled “Angiographically borderline left main coronary artery lesions: correlation of transthoracic Doppler echocardiography and intravascular ultrasound. A pilot study.” by Zoltán Ruzsa et al.

We would like to publish it in Cardiovascular Ultrasound Journal as an original article. We tried to answer all the questions arisen.

This original article highlight the importance of the transthoracic Doppler echocardiography in the detection of borderline left main coronary artery stenosis. The TTDE gives important physiological information about the functional severity of the borderline left main coronary artery stenosis.

The manuscript type is an Original Article and represents original work that is not being considered elsewhere for publication, in whole or in part. The authors have no real or perceived conflicts of interest to disclose. This manuscript in not a resubmission of a previously rejected paper nor is it a submission for a special issue. The manuscript has been read and approved by all the authors.

Major and minor errors were corrected in the final form of the publication.

Best regards,
Zoltán Ruzsa, MD and Albert Varga, MD, PhD, FESC